Author’s response to reviews

Title: The expression of sirtuins, superoxide dismutase, and lipid peroxidation status in peripheral blood from patients with diabetes and hypothyroidism

Authors:

Abdullah Alkhaldi (abmk2004@hotmail.com)

Samar Sultan (sasultan@kau.edu.sa)

Version: 1 Date: 25 Sep 2018

Author’s response to reviews:

Technical Comments:

Editor Comments:

1. We note that the current submission contains some textual overlap with other previously published works, in particular:


This overlap mainly exists in the methods and results sections of your manuscript.

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.

Please re-phrase these sections to minimise overlap.

Author response: This now has been re-phrased
2. Please change "Data deposition" to be "Availability of Data and Materials".

We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it in this section. If you do not wish to share your data, please clearly state this in this section along with a justification. Data availability statements can take one of the following forms (or a combination of more than one if required for multiple datasets):

- The datasets generated and/or analysed during the current study are available in the [NAME] repository, [PERSISTENT WEB LINK TO DATASETS]

- The datasets used and/or analysed during the current study available from the corresponding author on reasonable request.

- The datasets generated and/or analysed during the current study are not publicly available due [REASON WHY DATA ARE NOT PUBLIC] but are available from the corresponding author on reasonable request.

- The data that support the findings of this study are available from [third party name] but restrictions apply to the availability of these data, which were used under license for the current study, and so are not publicly available. Data are however available from the authors upon reasonable request and with permission of [third party name].

Author response: This now has been changed

3. Please remove the funding information from the Acknowledgements and include it in the Funding section instead. If you have no further acknowledgements please put “Not Applicable” in the Acknowledgements section.

Author response: This now has been removed
4. Please combine your ethical approval and consent statements under the heading "Ethical approval and consent to participate".

Please also clarify if the consent to participate was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

Author response: This now has been changed and clarified

5. We note that you have not included a ‘Consent for publication’ section in the Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.

Author response: This now has been added

6. Please change your “Introduction” heading to instead read “Background”

Author response: This now has been changed